Author's response to reviews

Title: Socio-demographic and AIDS-related factors associated with tuberculosis stigma in southern Thailand: a quantitative, cross-sectional study of stigma among patients with TB and healthy community members

Authors:

Aaron M Kipp (aaron.kipp@vanderbilt.edu)
Petchawan Pungrassami (petchawanp@yahoo.com)
Kittikorn Nilmanat (k_nilmanat@hotmail.com)
Sohini Sengupta (sengups@unc.edu)
Charles Poole (cpoole@unc.edu)
Ronald P Strauss (ron_strauss@unc.edu)
Virasakdi Chongsuvivatwong (cvirasak@medicine.psu.ac.th)
Annelies Van Rie (vanrie@email.unc.edu)

Version: 5 Date: 26 August 2011

Author's response to reviews:

Dear publication team,

Per your recent request, I have made three changes to the manuscript:

1) added the department for Kittikorn Nilmanat and
2) using superscripts to identify the authors' institutions
3) reformating the authors' email addresses

Sincerely,

Aaron Kipp